Author’s response to reviews

Title: Decision-making styles in the context of colorectal cancer screening

Authors:
Linda Douma (l.douma@vumc.nl)
Ellen Uiters (ellen.uiters@rivm.nl)
Danielle Timmermans (drm.timmermans@vumc.nl)

Version: 5 Date: 23 Jan 2020

Author’s response to reviews:

Response to editor comments

1. Comment 1:
In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Our response:
We have added a sentence about the role of the funding body.

Revision (Funding, lines 597-598, page 28):
The funding body had no role in the design of the study, in the collection, analysis, and interpretation of data and in writing the manuscript.

2. Comment 2:
In the supplementary material, please provide only the English version of the questionnaire.

Our response:
We have removed the Dutch version of both questionnaires (and, accordingly, adjusted the reference to it in the text (line 245)).

3. Comment 3:
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Our response:
We have uploaded the files as requested.